Author’s response to reviews

Title: Low expression of a few genes indicates good prognosis in estrogen receptor positive breast cancer

Authors:

   Steven Buechler (buechler.1@nd.edu)

Version: 4 Date: 16 July 2009

Author’s response to reviews: see over
July 16, 2009

Manuscript: 2100358013257501 - Low expression of a few genes indicates good prognosis in estrogen receptor positive breast cancer.

To the editors of BMC Cancer:

The attached revised manuscript responds to the required changes in format. Point-by-point changes are as follows.

1. The manuscript was reformatted to a single column width throughout.
2. The Author’s contributions section was completed, even though I am the sole author.
3. In references 16, 17 and 18 the volume, issue and page numbers were added to each.
4. The country affiliation was added to my address.
5. Extraneous vertical and horizontal lines were removed from all tables.
6. In Table 2 the second sentence in the title block was moved to a legend below the table.
7. Several typographical errors were fixed.

Respectfully submitted,

Steven Buechler
Department of Mathematics
255 Hurley Hall
University of Notre Dame
Notre Dame, IN 46556

Phone: 574-302-5396
Email: buechler.1@nd.edu